Author’s response to reviews

Title: The effects of Vitamin C supplementation on pre-eclampsia in Mulago Hospital, Kampala, Uganda: a Randomized Placebo controlled clinical trial

Authors:

Paul Kiondo Dr (kiondop@yahoo.com)
Gakenia Wamuyu-Maina Dr (wmaina@rcqhc.org)
Julius Wandabwa Dr (giwandabwa@yahoo.com)
Gabriel S Bimenya Prof (gsbimenya@chs.mak.ac.ug)
Nazarius M Tumwesigye Dr (nmtumwesigye@gmail.com)
Pius Okong Prof (mrokong@gmail.com)

Version: 3 Date: 11 September 2013

Author’s response to reviews: see over
Dear Sir/Madam

RE: Submission of an article for publication

I would like to submit a clinical article entitled “To evaluate the effects of Vitamin C supplementation on pre-eclampsia in Mulago Hospital, Kampala, Uganda” to your Journal for publication.

This work was conducted in Mulago Hospital from November 2011 to January 2013 in the Department of Obstetrics and Gynecology. The management of the Hospital has granted us permission to publish this work in your Journal.

This article has not been submitted in any other journal for publication. There is also no conflict of interest to be declared by the authors.

I would like to state that this article has been read by all the authors and it represents the work which was conducted in the Hospital by the authors. I would like to state further that all the requirements of the authors have been met.

The following changes have been made.

   i. The word CONSORT is mentioned in the manuscript and the CONSORT check list has been included in the submission.

   ii. The title page has been included in front of the manuscript.

   iii. The tables have been formatted using the word processor.

Yours faithfully

Dr Paul Kiondo
Consultant Obstetrician/Gynaecologist
Senior Lecturer, College of Health Sciences
P.O Box 7072, Kampala, Uganda

Email kiondop@yahoo.com

Corresponding author.

Telephone Number (mobile) +256 772 501937